Author’s response to reviews

Title: Low trust of ethnic minority patients in in-hospital care: a qualitative study

Authors:

Jeanine Suurmond (j.suurmond@amc.uva.nl)
Ellen Uiters (ellenuiters@rivm.nl)
Martine C. de Bruijne (mc.debruijne@vumc.nl)
Karien Stronks (k.stronks@amc.uva.nl)
Marie-Louise Essink-Bot (M.L.Essink-Bot@amc.uva.nl)

Version: 2 Date: 19 February 2010

Author’s response to reviews:

15 February 2010

Dear Madam/Sir,

Please find attached our paper “Low trust of ethnic minority patients after in-hospital care: a qualitative study”. To our knowledge this is one of the first studies that gives a thorough insight in the different factors that contribute to ethnic minority patients having low trust in care providers in the hospital. Insight in these factors may contribute to the decrease of ethnic disparities in health.

We look forward receiving your reply.

Kind regards,